Author's response to reviews

Title: Obesity and Hypertension in an Iranian Cohort Study; Iranian Women Experience Higher Rates of Obesity and Hypertension Than American Women

Authors:

Hossein H Bahrami Dr (hbahrami@jhsph.edu)
Mohsen M Sadatsafavi Dr (Safavi@ddrcir.org)
Akram A Pourshams Dr (Pourshams@ams.ac.ir)
Farin F Kamangar Dr (kamangaf@mail.nih.gov)
Mehdi M Nouraie Dr (nouraie@ddrcir.org)
Shahriar S Semnani Dr (sh_semnani@yahoo.com)
Paul P Brennan Dr (brennan@iarc.fr)
Paolo P Boffetta Dr (boffetta@iarc.fr)
Reza R Malekzadeh prof (malek@ams.ac.ir)

Version: 2 Date: 21 March 2006

Author's response to reviews:

some changes to the formatting of manuscript